Author's response to reviews

Title: "Men's health - a little in the shadow": a formative evaluation of medical curriculum enhancement with men's health teaching and learning.

Authors:

Carol A Holden (carol.holden@monash.edu)
Veronica R Collins (veronica.collins@monash.edu)
Christopher J Anderson (christopher.anderson@monash.edu)
Sylvia Pomeroy (sylia.pomeroy@monash.edu)
Richard Turner (richard.turner@utas.edu.au)
Benedict J Canny (ben.canny@monash.edu)
Bu B Yeap (byeap@cyllene.uwa.edu.au)
Gary Wittert (gary.wittert@adelaide.edu.au)
Robert I McLachlan (robert.mclachlan@mimr-phi.org)

Version: 2
Date: 12 May 2015

Author's response to reviews: see over
12\textsuperscript{th} May 2015

\textbf{Aldrin Joseph Ulep}
Journal Editorial Office
\textbf{BioMed Central}

\textbf{Re: MS: 1423545991673965}

\textbf{Research article: A community of practice approach for curriculum enhancement: evaluation of a men’s health framework as a case study.}

BMC Medical Education (Section: Curriculum development)

Dear Aldrin

Many thanks for your prompt review of our submitted manuscript and your editorial suggestions.

We have incorporated your helpful suggestions by revising the manuscript as follows:

- Changed the title of the manuscript to clarify the major focus of the study. It is now: \textit{“Men’s health – a little in the shadow”: a formative evaluation of medical curriculum enhancement with men’s health teaching and learning.} Could you please amend your recordings to reflect this changed title.
- Included further information (lines 377 – 394) about the evaluation of the proposed CoP, and
- Reduced the number of quotes by removing repetitive or unnecessary quotations.

We have also made some minor editorial amendments to the manuscript text.

Thank you again for your feedback and your consideration of this article for publication in BMC Medical Education.

Yours sincerely,

\begin{flushright}
Carol Holden PhD, MPH \\
CEO, Andrology Australia \\
Monash University
\end{flushright}